Author's response to reviews

**Title:** CD209 in inflammatory bowel disease: a case-control study in the Spanish population

**Authors:**

Concepcion Nunez (conchita.npardo@gmail.com)
Javier Oliver (javi_om@ipb.csic.es)
Juan Luis Mendoza (jmendoza@meditex.es)
Maria Gomez-Garcia (mmrosarioes@yahoo.es)
Carlos Taxonera (ctaxo@inicia.es)
Luis M. Gomez (lgomez@cib.org.co)
Miguel A. Lopez-Nevot (nevot@juntadeandalucia.es)
Emilio G. de la Concha (egomezdela.hcsc@salud.madrid.org)
Elena Urcelay (eurcelay.hcsc@salud.madrid.org)
Alfonso Martinez (alfmdoncel@gmail.com)
Javier Martin (martin@ipb.csic.es)

**Version:** 4 **Date:** 3 December 2007

**Author's response to reviews:** see over
CD209 IN INFLAMMATORY BOWEL DISEASE: A CASE-CONTROL STUDY IN THE SPANISH POPULATION

Dear editor,

We have read the editor comments and performed the changes in the manuscript according to them. In the following point by point description we detail the changes performed.

Point by point description:

1. We have removed the authors' qualifications from the title page.
2. We have included the country at the end of the affiliation details.
3. We have written Authors' contributions instead of Author's contribution.
4. Tables 2 and 3 were not correctly numbered in the text. We have performed the pertinent changes and now all tables are well referenced within the text and they are in numerical order.

Yours sincerely,

A. Martínez

Alfonso Martínez, Servicio de Inmunología Clínica, Hospital Clínico San Carlos. Address: C/ Martín Lagos s/n. 28040 Madrid, Spain. E-mail: alfmdoncel@gmail.com